Author's response to reviews

Title: Use of local anaesthetics and adjuncts for spinal and epidural anaesthesia and analgesia at German and Austrian University Hospitals: An online survey to assess current standard practice

Authors:

Bianca M Wahlen (wahlen_b@klinik.uni-wuerzburg.de)
Norbert Roewer (Anaesthesie-Direktion@klinik.uni-wuerzburg.de)
Peter Kranke (Kranke_P@klinik.uni-wuerzburg.de)

Version: 6 Date: 28 March 2010

Author's response to reviews:

Dear ladies and gentlemen

Thanks a lot for your e-mail. The following changes have been made.
1. In "Authors`contributions" "All authors have read and approved the manuscript" is added.
2. The uploaded tables have been removed from the submission system and have been included within the text file of the manuscript following the references. The table title has been moved to above the table and the legend has been moved to below the table within the text.

You mentioned, that the tables should not be embedded images or drawings and should be formatted from cells, not tabbed text. As we created the tables via "Word" and did not use tabbed text, I am not sure what you refer to. If there are any further changes to be made concerning the tables or you want us to use "Excel" please let me know.
3. "Fig." in the text has been changed to "Figure" in all four cases
4. Some typographical errors have been corrected.

Thanks a lot in advance.
I hope to hear from you soon.
Sincerely Yours
Bianca Maria Wahlen, MD